Author’s response to reviews

Title: Therapeutic effect of intravesical administration of paclitaxel solubilized with poly(2-methacryloyloxyethyl phosphorylcholine-co-n-butyl methacrylate) in an orthotopic bladder cancer model

Authors:

Koetsu Tamura (kohetu@yahoo.co.jp)
Eiji Kikuchi (eiji-k@kb3.so-net.ne.jp)
Tomohiro Konno (konno@bioeng.t.u-tokyo.ac.jp)
Kazuhiko Ishihara (ishihara@mpc.t.u-tokyo.ac.jp)
Akira Miyajima (akiram@sc.itc.keio.ac.jp)
Mototsugu Oya (moto-oya@sc.itc.keio.ac.jp)

Version: 2
Date: 21 September 2014

Author’s response to reviews: see over
Dr. Ryan M. Relox  
Editorial Office ‘BMC cancer’

Dear Dr. Relox

Thank you for considering our paper “Therapeutic effect of intravesical administration of paclitaxel solubilized with poly (2-methacryloyloxyethyl phosphorylcholine-co-n-butyl methacrylate) in an orthotopic bladder cancer model” (MS: 1028495963139995) for publication in BMC Cancer. All changes have been highlighted in the revised paper and response to all comments follows below.

Sincerely,

Koetsu Tamura, MD.  
Department of Urology,  
Keio University School of Medicine.  
35 Shinanomachi, Shinjuku-ku, Tokyo 160-8582, Japan.  
Phone: 81-3-5363-3825 Fax: 81-3-3225-1985  
E-mail: kohetu@yahoo.co.jp

1) Requesting line numbering.  
Line numbers have been added.

2) Please provide the full name of the animal ethics committee that approved the study.  
The full name of the animal ethics committee has been provided, page 8 line 8: All of the procedures involving animals and their care in this study were approved by the Animal Care Committee of Keio University in accordance with institutional and Japanese government guidelines for animal experiments.

3) Requesting list of abbreviations.  
List of abbreviations have been provided, page 16 line 11:  

List of Abbreviations  
LC-MS/MS liquid chromatography tandem mass spectrometry  
PBS phosphate buffered saline  
LDH lactate dehydrogenase